Author's response to reviews

Title: Patterns of Condom use and associated factors among HIV positive clients in North Western Ethiopia: A comparative cross sectional study

Authors:

Estifanos Yalew BSc, MPH (e.estifon@yahoo.com)
Desalegn T zegeye MD, MPH (zdesalegn@gmail.com)
Solomon Meseret BSc, MPH (messisol@yahoo.com)

Version: 2  Date: 23 September 2011

Author's response to reviews:

Hello sir/madam

Well come!

Based on the editorial comment on the method section regarding selection of study participants, the authors believe it and tried to change as follows:

1. We made some changes on the title of research article by adding the particular study design used for this particular study.

2. In the method section the authors tried to clarify more on how the study participants were selected. They rewrite just like this: participants were selected using systematic random sampling technique based on their daily visits to the outpatient department of ART clinic during the study period from both ART experienced and naïve clients.

3. The author more tried to show on how and where pretest was done as well minor spelling/grammar editorial were done.

Thanks for your patience and understanding.

With best regards!
the authors